Author's response to reviews

Title: An increase in dietary n-3 fatty acids decreases a marker of bone resorption in humans

Authors:

Amy E Griel (aeg126@psu.edu)
Penny M Kris-Etherton (pmk3@psu.edu)
Kirsten F Hilpert (kfc107@psu.edu)
Guixiang Zhao (gzhao5@yahoo.com)
Sheila G West (sgw2@psu.edu)
Rebecca L Corwin (rxc13@psu.edu)

Version: 6 Date: 5 January 2007

Author's response to reviews:

I have made all of the requested formatting changes.

Thank you,
Amy Griel